Author's response to reviews

Title: AhR transcriptional activity in serum of Inuit's across Greenlandic districts

Authors:

Manhai Long (ml@mil.au.dk)
Bente Deutch (bd@mil.au.dk)
Eva C Bonefeld-Jorgensen (ebj@mil.au.dk)

Version: 7 Date: 11 October 2007

Author's response to reviews:

Cover letter

- The abstract in the manuscript is matched with that in submission system.
- The references are formatted using EndNote with BioMed Central style.
- The changes are done in the EndNote.
- Large groups of references e.g. [2-5] have been changed to [2] [3] [4] [5].